Author’s response to reviews

Title: Changes in Diet and Physical Activity Resulting from the Shape Up Somerville Community Intervention

Authors:

Sara C Folta (sara.folta@tufts.edu)
Julia F Kuder (julia.kuder@gmail.com)
Jeanne P Goldberg (jeanne.goldberg@tufts.edu)
Raymond R Hyatt (raymond.hyatt@tufts.edu)
Aviva Must (aviva.must@tufts.edu)
Elena Naumova (elena.naumova@tufts.edu)
Miriam E Nelson (miriam.nelson@tufts.edu)
Christina D Economos (christina.economos@tufts.edu)

Version: 2 Date: 18 February 2013

Author’s response to reviews: see over
18 February 2013

Dear BMC Pediatrics Editorial Board:

Accompanying this letter please find a revised version of our article, “Changes in Diet and Physical Activity Resulting from the Shape Up Somerville Community Intervention”, manuscript #7610741489151498 which we are submitting as a Research Article. We have responded to the initial format review by including a title page and abstract in the document with the main manuscript. We have also included the clinical trial number (NCT00153322) for the study. We regret these omissions from the original submission.

Sincerely,

Sara C. Folta, PhD
Corresponding Author

Julia F. Kuder, MA
Jeanne P. Goldberg, PhD, RD
Raymond R. Hyatt, PhD
Aviva Must, PhD
Elena Naumova, PhD
Miriam E. Nelson, PhD
Christina D. Economos, PhD